**Author's response to reviews**

**Title:** Clinical efficacy and prognostic indicators for lower limb pedalling exercise early after stroke: A pilot randomised controlled trial

**Authors:**

Nicola J Hancock (n.hancock@uea.ac.uk)
Lee Shepstone (l.shepstone@uea.ac.uk)
Philip Rowe (philip.rowe@strath.ac.uk)
Phyo K Myint (phyo.k.myint@uea.ac.uk)
Valerie Pomeroy (v.pomeroy@uea.ac.uk)

**Version:** 3  **Date:** 25 February 2011

**Author's response to reviews:** see over
To: Editors-in-Chief: Doug Altman, Curt Furberg, Jeremy Grimshaw and Peter Rothwell  
Reviewer: David Kent  
trialsjournal.com  

26th February 2011  

Re: Clinical efficacy and prognostic indicators for lower limb pedalling exercise early after stroke: Study protocol for a pilot randomised controlled trial; MS: 1757957420466908  

Thank you for the notification on 22.02.11 that our manuscript is close to formal acceptance by the Editorial board. We were delighted to receive this news.  

As requested, the manuscript has been amended; 1) to include “study protocol”  

Yours Sincerely  

Nicola J Hancock  

Nicola J Hancock  
Chief Investigator  

Associate Tutor/Post-graduate Researcher  
Institute of Health and Social Sciences  
The Queen’s Building  
University of East Anglia  
Norwich NR4 7TJ